Author's response to reviews

Title: Pesticide use and opportunities of exposure among farmers and their families: cross-sectional studies 1998-2006 from Hebron governorate, occupied Palestinian territory

Authors:

Yaser Issa (yaser_74@hotmail.com)
Farid Abu Sham’a (farid.shama@yahoo.com)
Khaldoun Nijem (knijem@hotmail.com)
Espen Bjertness (espen.bjertness@medisin.uio.no)
Petter Kristensen (petter.kristensen@stami.no)

Version: 5 Date: 18 October 2010

Author's response to reviews:

Dear Editor-in-Chief

Environmental Health

#

We have done all your final comments and made the following changes:
- all references in the text now are consecutive and presented in numerical order.
- we removed some references were existed in the reference list and not mentioned in the text. These references were existed due to earlier revision and forgotten to be removed. Now all references in the text and in the list fit to each other.